### **UC Davis**

### **UC Davis Previously Published Works**

#### **Title**

Potential for community based surveillance of febrile diseases: Feasibility of selfadministered rapid diagnostic tests in Iquitos, Peru and Phnom Penh, Cambodia

#### **Permalink**

https://escholarship.org/uc/item/4gk0f2dv

#### **Journal**

PLOS Neglected Tropical Diseases, 15(4)

#### **ISSN**

1935-2727

#### **Authors**

Morrison, Amy C Schwarz, Julia Mckenney, Jennie L et al.

#### **Publication Date**

2021

#### DOI

10.1371/journal.pntd.0009307

Peer reviewed



#### G OPEN ACCESS

Citation: Morrison AC, Schwarz J, Mckenney JL, Cordova J, Rios JE, Quiroz WL, et al. (2021) Potential for community based surveillance of febrile diseases: Feasibility of self-administered rapid diagnostic tests in Iquitos, Peru and Phnom Penh, Cambodia. PLoS Negl Trop Dis 15(4): e0009307. https://doi.org/10.1371/journal.pntd.0009307

Editor: David Joseph Diemert, George Washington University School of Medicine and Health Sciences, UNITED STATES

Received: June 16, 2020
Accepted: March 12, 2021
Published: April 26, 2021

Copyright: This is an open access article, free of all copyright, and may be freely reproduced, distributed, transmitted, modified, built upon, or otherwise used by anyone for any lawful purpose. The work is made available under the Creative Commons CCO public domain dedication.

**Data Availability Statement:** All relevant data are within the manuscript and its Supporting Information files.

Funding: This research was supported by funding from the U.S. Department of Defense's Defense Threat Reduction Agency (DTRA), the U.S. National Institutes of Health, the U.S. National Institute of Allergy and Infectious Diseases (NIH/NIAID) award

RESEARCH ARTICLE

## Potential for community based surveillance of febrile diseases: Feasibility of selfadministered rapid diagnostic tests in Iquitos, Peru and Phnom Penh, Cambodia

Amy C. Morrison 1,2\*, Julia Schwarz 3, Jennie L. Mckenney, Jhonny Cordova, Jennifer E. Rios, W. Lorena Quiroz, S. Alfonso Vizcarra, Heng Sopheab 6, Karin M. Bauer 5,8, Chhorvann Chhea, Vonthanak Saphonn, Robert D. Hontz, Pamina M. Gorbach 4, Valerie A. Paz-Soldan

- Department of Pathology, Microbiology, and Immunology, School of Veterinary Medicine, University of California, Davis, California, United States of America,
   U.S. Naval Medical Research Unit No. 6 (NAMRU-6), Lima, Peru,
   Icahn School of Medicine at Mt Sinai, New York, New York, United States of America,
   University of California Fielding School of Public Health, Los Angeles, California, United States of America,
   Department of Entomology and Nematology, University of California, Davis, California, United States of America,
   School of Public Health, National Institute of Public Health, Phnom Penh, Cambodia,
   Tulane School of Public Health and Tropical Medicine, New Orleans, Lousiana, United States of America,
   University of Washington, Seattle, Washington, United States of America,
   University of Mashington, Seattle, Washington, United States of America,
   University of Mashington, Seattle, Washington, United States of America,
   University of Health Sciences, Phnom Penh, Cambodia,
   U.S. Naval Medical Research Unit No. 2 (NAMRU-2), Singapore
- \* amy.aegypti@gmail.com, amy.c.morrison3.civ@mail.mil

#### **Abstract**

Rapid diagnostic tests (RDTs) have the potential to identify infectious diseases quickly, minimize disease transmission, and could complement and improve surveillance and control of infectious and vector-borne diseases during outbreaks. The U.S. Defense Threat Reduction Agency's Joint Science and Technology Office (DTRA-JSTO) program set out to develop novel point-of-need RDTs for infectious diseases and deploy them for home use with no training. The aim of this formative study was to address two questions: 1) could community members in Iguitos, Peru and Phnom Penh, Cambodia competently use RDTs of different levels of complexity at home with visually based instructions provided, and 2) if an RDT were provided at no cost, would it be used at home if family members displayed febrile symptoms? Test kits with written and video (Peru only) instructions were provided to community members (Peru [n = 202]; Cambodia [n = 50]) or community health workers (Cambodia [n = 45]), and trained observers evaluated the competency level for each of the several steps required to successfully operate one of two multiplex RDTs on themselves or other consenting participant (i.e., family member). In Iquitos, >80% of residents were able to perform 11/12 steps and 7/15 steps for the two- and five-pathogen test, respectively. Competency in Phnom Penh never reached 80% for any of the 12 or 15 steps for either test; the percentage of participants able to perform a step ranged from 26–76% and 23–72%, for the two- and five-pathogen tests, respectively. Commercially available NS1 dengue rapid tests were distributed, at no cost, to households with confirmed exposure to dengue or Zika virus; of 14 febrile cases reported, six used the provided RDT. Our findings support the need for

number P01 Al098670 (ACM, VAP co-leaders), the Armed Forces Health Surveillance Division, Global Emerging Infections Surveillance Branch (GEIS) ProMIS ID: P0106\_18\_N6\_01.01, and the Military Infectious Disease Research Program (S0263\_10\_LI and S0216\_09\_LI to RDH). The funders had no role in the study design, data collection and analysis, decision to publish or preparation of the manuscript.

**Competing interests:** The authors have declared that no competing interests exist.

further implementation research on the appropriate level of instructions or training needed for diverse devices in different settings, as well as how to best integrate RDTs into existing local public health and disease surveillance programs at a large scale.

#### Author summary

Rapid diagnostic tests (RDTs) are becoming increasingly available and are useful during outbreaks of infectious and vector-borne diseases. This study set out to answer two questions regarding the use of RDTs in community settings. First, how well could community members at two diverse sites-Iquitos, Peru and Phnom Penh, Cambodia-perform an RDT on themselves and willing family members with only written and/or video instructions, and with two tests of different complexity? Second, and only in Iquitos, if RDTs were made widely available at no cost to households, would residents choose to use one for a family member showing febrile symptoms? In Peru, 202 community members were provided written and/or video instructions and a test kit, and then were observed operating the RDT on themselves or family members in their own homes. In Cambodia, 45 Community Health Workers and 50 community members were provided only written instructions at a local health center and asked to carry out the test. In both locations, trained observers scored each of the 12-15 steps necessary to conduct the test on a 3-point scale (1 being lowest proficiency level, 3 being highest), for one of the two RDTs in this study. In Iquitos, >80% of residents were able to perform 11/12 steps and 7/15 steps for the two- and five-pathogen test, respectively. Competency in Phnom Penh never reached 80% for any of the 12 or 15 steps for either test, with the percentage of participants demonstrating high competency ranging from 26-76% and 23-72%, for the two- and fivepathogen tests, respectively. Finally, in Iquitos, RDTs were distributed to households with earlier confirmed cases of dengue or Zika virus: about half of all household members that developed fevers were self-tested or tested by another family member. This study is consistent with our earlier findings from focus group discussions suggesting that implementation of simple home-based rapid diagnostic devices, could be used in different settings, but further research is needed to determine the best type and level of instructions to do this.

#### Introduction

As rapid diagnostic tests (RDTs) become easier to use and increasingly available, their application for early detection of disease, especially outside traditional point-of-care settings, has considerable potential for use in disease surveillance programs; for example, options for self-testing during the COVID-19 pandemic. RDTs available for home use have been of particular benefit for blood glucose monitoring in diabetics [1], pregnancy testing [2], and detection of infections associated with social stigmas, such as sexually transmitted infections (HIV, syphilis, hepatitis) [3]. RDTs are also extremely useful for identifying infectious diseases in remote settings without complex equipment, but the potential for their use in community-based or sentinel surveillance programs or incorporation into national treatment programs for diseases other than malaria is largely untapped [4].

Unlike pregnancy tests or in-home HIV tests, which only require dipping a stick in urine or swabbing a gum, RDTs for dengue or other febrile illnesses involve more complex steps such

as using a lancet to obtain blood, mixing buffers, and applying appropriate mixtures to the testing device. These latter steps were the primary obstacles for successful use of home-based HIV and malaria RDTs [5–8]. Initially, RDTs for self-monitoring of blood glucose also encountered these problems, but early studies led to test improvements, and these RDTs are now widely used and highly accurate [1].

If reliable and able to be used only with instructions (i.e., no training), home-based RDTs could be used for screening and/or surveillance of both dengue virus, SARS-CoV-2, or other infectious diseases to reduce transmission risk associated with seeking care at health facilities. The impact of home- or community-based RDT use on health systems will depend on validation that untrained community members or lay health workers can use them effectively. The consequences, such as not seeking needed health care for a false negative result or a negative result associated with the timing of the test (i.e., for dengue, too early for the detection of antibodies or too late for the detection of NS1), require consideration prior to widespread use [9]. Similarly, people testing positive must have information on how and where to seek appropriate clinical care for follow up.

In 2012, the U.S. Defense Threat Reduction Agency Joint Science and Technology Office (DTRA-JSTO) initiated a program to develop novel point-of-need diagnostics for surveillance of emerging infectious diseases that included, beyond evaluation of the sensitivity and specificity of the RDTs, a component to assess acceptability and competency for use. Our research carried out under this program was formative and focused on supporting testing of multiplex devices in the DTRA pipeline that included important bio-threats such as plague and melioidosis. Specifically, two RDTs were used in this formative research study: one that detected two pathogens (dengue and Burkholderia pseudomallei) and the other that detected five pathogens (dengue, B. pseudomallei, Plasmodium vivax and falciparum, Yersinia pestis). In February 2014, focus group discussions with community members and health professionals conducted in Iquitos, Peru and Phnom Penh, Cambodia showed that there was acceptability for at-home use of RDTs in Peru but less enthusiasm in Cambodia, and that a principal motivator for RDT use would be a link to a streamlined process for clinical feedback or follow up based on the test result [9]. There was a strong preference for RDTs to be applied by community health workers (CHWs) in Cambodia, where CHW infrastructure for malaria control was in place in our study communities [9]. Findings from the focus groups were used to guide this next phase of research.

Our objective was to answer the following two questions related to RDT use in dengue endemic settings, represented by Cambodia and Peru. First, could community members (both sites) and CHWs (only Phnom Penh) with appropriate instructions (written and/or video) competently use RDTs and did competency vary with test complexity? Second, would community members (Peru only) use RDTs to self-administer or test a family member if they displayed febrile symptoms? This formative research study was not meant to be generalizable; however, the findings and implications related to this study have broader applications.

#### Methods and materials

#### **Ethics statement**

For work in Peru, the study protocol was approved by the U.S. Naval Medical Research Unit No. 6 (Protocol #NAMRU6.2014.0003) Institutional Review Board (IRB), which includes Peruvian representation and complies with U.S. Federal and Peruvian regulations governing the protection of human subjects. IRB Reliance Agreements were established between U.S. Naval Medical Research Unit No. 6 (NAMRU-6), and the University of California, Davis, as well as Tulane University School of Public Health and Tropical Medicine for the NAMRU-6

IRB to be the reviewing IRB. Additionally, the protocol was reviewed and approved by the Loreto Regional Health Department, which oversees health research in Iquitos. For work in Cambodia, the study protocol was approved by the National Ethics Committee in Cambodia and the University of California, Los Angeles IRB. For the first research question involving participant observation while applying the RDT to assess competency with tests of different complexity, written consent was obtained from all adult participants (≥18 years of age) that were either self-applying the RDT, or having another participant administer the RDT on them. Children who participated in the role of "patient" provided written assent (8–17 years of age) and a parent or guardian provided written parental consent for children age 2–17. For the second research question, RDTs were provided to higher-risk households to assess whether people would choose to use them ("RDT adoption" from now on) on a febrile household member. All eligible family members (≥2 years of age) provided written consent to self-administer the RDT or have a family member administer the RDT if they displayed febrile symptoms.

#### Study setting

This study took place in Iquitos, Peru, and Phnom Penh, Cambodia, described in detail previously [9,10]. Iquitos is a large urban center (population ~ 400,000) where residents have easy access to a network of government health facilities and hospitals, but they do not have a network of health workers within the city. Although malaria is endemic in villages outside Iquitos, it does not circulate inside the city [11]. At the time of this competency observation study, the Asian-American genotype of dengue virus serotype 2 was circulating and Zika virus had not been detected in the city [12]. Study subjects were recruited between May 2015 and March 2016 from ten city blocks selected at random from four neighborhoods, two with ongoing prospective cohort studies [13-15] and two with no exposure to human research studies to control for the possibility that prior exposure as a research participant might improve competency. Within these randomly selected blocks, our research team recruited participants door-to-door, with one adult per household, until we reached a target of 50 participants per neighborhood. When a participant agreed to participate, our staff would call an administrator who provided the assignment of the two- or five- line device from a randomized list. Related to the question of RDT use adoption, Iquitos has ongoing cohort studies for dengue virus that monitor residents from houses in study neighborhoods three times per week for febrile illness, with subsequent laboratory confirmation if that illness is due to dengue or Zika virus [16,17]. The latter triggers contact cluster investigations where laboratory confirmed cases are interviewed to identify residential locations where the individual spent time in the 15-days prior to infection [14,18,19]. All residents in those locations are then monitored for secondary cases. We leveraged our ongoing cluster investigations to assess RDT use adoption (research question #2) in Iquitos between September 2016-September 2018.

The Cambodian sites were peri-urban areas on the outskirts of Phnom Penh with an active community health worker (CHW) program. CHWs were integrated in both dengue and malaria surveillance programs and provided basic health care to neighborhood residents. During our previous study examining acceptability of RDT use at home, participants revealed a preference for having CHWs apply RDTs in their communities [9]. As a result, both community members and CHWs were selected by convenience sampling, then invited to a local health center to apply the RDTs on others, and their competency was assessed by our Cambodian team.

#### Rapid diagnostic devices and application

**RDT use competency.** The two prototype devices used in the observation study were under evaluation by DTRA [20], whose long-term goals were to develop multiplex devices to

detect a wide array of pathogens including Biological Safety Level III/IV agents that could be applied directly by community members. Neither device was developed further, but the DTRA-funded observation studies were for these devices, which provided us a model system to test devices of varied complexity. The first RDT identified dengue virus and *Burkholderia pseudomallei*, henceforth called the two-line test; and the second identified five pathogens—dengue virus, *Plasmodium vivax/falciparum*, *Yersinia pestis*, *B. pseudomallei*—henceforth called the five-line test. Both devices required collection of whole blood via a finger-stick. The two-line test required 12 steps to complete, whereas the five-line test had 15 steps. As both prototype devices were still being evaluated, and not approved for diagnostic use, the research assistants explained to the participants that there were no meaningful results from these RDTs.

**RDT Use adoption.** For this objective, we used an RDT that was commercially available, the SD Bioline DENGUE Duo ("NS1"), which also performed up to label expectations in a multi-country comparison study [21]. We selected this device because it was approved for use in Peru, providing a diagnostic result to the user—the principal motivator for use. We developed and piloted pictorial and written instructions, an instructional video, and a result reporting form distributed with the device (S1 and S2 Texts).

#### Instructional materials

In both locations, detailed systematic instructions with pictures or drawings were developed for low literacy users (S1 Text), piloted in the community prior to use during FGDs, and finalized in both sites—one in real time during the first phase (Iquitos) and one right after the first phase (Phnom Penh)—prior to their application in the second phase. Based on overly complicated instructions that came with the devices, the Peruvian field team developed a one-page document consisting of images and minimal text descriptions of each step. These initial instruction drafts were piloted among the local research assistants, staff, and community members, followed by feedback from FGD participants, who were able try using RDT [9]. The lead scientists from the Peruvian research team (VPS, ACM) led the FGDs, and based on participant feedback and their previous experience producing video instructions [10,22], the potential value of video instructions was recognized; a video was immediately developed and validated in the subsequent FGDs. A DVD containing the final version of the video was included with the test kits to observation study participants (S2 Text. https://youtu.be/ 5RQtc9KNA5A, https://youtu.be/adv-yQaI iA, https://youtu.be/arOSOaeulIw). In Cambodia, draft instructions and graphic images were developed from the manufacturer's instructions, translated to Khmer, piloted with local community members and health center staff prior to their use in FGDs. In contrast to Peru, RDT use by FGD participants was not approved by local IRBs, so adaptation materials in real-time and video development was not possible; instead FGDs used the original piloted written materials.

#### Study procedures

Objective 1: Could community members in Iquitos and Phnom Penh competently use RDTs of different complexity at home without training?. The rationale for this study was to assess competency of community members or CHWs to apply RDTs without training beyond the written or video instructions provided with the device. After providing consent, participants were given a packet containing one of the devices, all other materials needed to complete testing (including retractable lancets and basic biosafety materials), and instructional materials (written one-page document, and in Iquitos, a DVD with video instructions). Participants were asked to follow the instructions in the packet, while a research assistant, using an observation form (S3 Text), evaluated and scored, on a 3-point scale (1 lowest proficiency, 3

highest), each step in the participants' use of the device (i.e., lay out materials before starting, determine which finger to prick, use lancet correctly). The observers (research assistants) piloted the observation form among themselves, co-workers (nurse technicians and health center staff), and family members, followed by training sessions where all observers watched the same individual carry out the test, compared and discussed scoring with the lead investigator (VPS), until consistent scoring was achieved among the observers. The research assistants did not interfere or assist the participant unless it was obvious to the observer that an imminent mistake affecting all subsequent steps was about to take place, or if a participant requested help. All assistance was documented.

In Phnom Penh, where FGDs indicated a preference for CHWs applying RDTs over community members, a small trial to evaluate the value of experience or practice for competency was carried out by observing eight CHWs applying the RDT up to six times to determine how long it would take to obtain at least 80% competency. The observers did not interfere or assist the CHWs unless it was obvious that a mistake affecting subsequent steps was about to happen, or if help was requested. That is, real-time feedback was provided between repeat tests only as needed.

# Objective 2: Would community members use RDTs to test their own or family member's febrile illness? (Peru only)

Participants responding to a hypothetical question about the option to use an RDT for febrile illness are likely to respond with what they perceive to be socially desirable. Hence, to determine whether community members would use an RDT in case of febrile illness, we decided to distribute commercially available dengue RDTs to families perceived to be at high risk for dengue or Zika based on proximity to other confirmed dengue or Zika cases in their communities. Leveraging the research infrastructure in Iquitos, where our research team has been conducting contact cluster investigations since 2008 [14], we identified laboratory-confirmed dengue or Zika cases ("index cases"), then carried out an interview to find out what locations the infected person had visited during the 15 days prior to becoming ill [23]. We then visited the index case's home and contact locations, and provided a kit with two RDTs (SD Bioline Dengue Duo), written and video instructions (\$1 and \$2 Texts), and all additional materials needed to carry out the test to each consenting household in the cluster. The kits were removed from each house after a 30-day follow up period if not used. During the follow up period, each household was visited every 1-2 days to identify new febrile cases which were tested by PCR. Our team carried out confirmatory testing if the person used the RDT. We recorded how many individuals showed symptoms of fever, whether they used the RDT, and if not, explored their rationale for not using the RDT.

#### **Analysis**

Research assistants provided a score, ranging from one (unable to complete the procedure without asking for help) to three (able to complete the step correctly without assistance) for *each individual step* needed to complete the test (12 steps were scored). An intermediate score of two was given when the participant demonstrated some degree of competency but needed some assistance or reassurance. To assess factors contributing to correct use of the RDT, we developed an overall "competency score" for each participant's use of one of the two RDTs. The scores for each of the 12 steps that both tests had in common were summed for an overall competency score (range 12–36) and divided by the 12 common steps evaluated (range 1–3). A binary response variable of > 2.9 defined as "highly competent" was used to construct separate logistic regression models for Peru and Cambodia. All analyses were conducted in SAS/STAT software, version 9.4 of the SAS system for Windows (SAS Institute, Cary, NC).

#### Results

#### Participant characteristics

The demographic characteristics of the study participants who applied an RDT are summarized in Table 1. In Peru, only community members were enrolled in the study; 203 participants were originally observed using an RDT, and one observation was excluded because the participant was a NAMRU-6 nurse technician with extensive experience using RDTs. In Cambodia, due to a stated preference for relying on CHWs, we observed RDTs being applied by both CHWs (n = 45) and community members (n = 50). In Iquitos, females comprised a significantly higher percentage of the participants and those recruited were significantly younger than in Phnom Penh (gender:  $\chi$ 2 = 4.5, df = 1; p = 0.0349; age:  $\chi$ 2 = 26.5, df = 1 p<0.0001).

Table 1. Summary of study population of device application by community members (excluding the 8 Community Health Workers in Cambodia that participated in a separate experiment described below).

Characteristic	Options	Peru	Cambodia
Participant	Community Health Worker	-	45 (47.4%)
	Household Resident	202 (100%)	50 (52.6%)
RDT	Two Line Test (12-steps)	101 (50.0%)	42 (44.2%)
	Five Line Test (15-steps)	101 (50.0%)	53 (55.8%)
Gender*	Females	142 (70.3%)	55 (57.9%)
	Males	60 (29.7%)	40 (42.1%)
Age (years)**	18–29	84 (41.6%)	24 (25.3%)
	30–39	49 (24.3%)	11 (11.6%)
	40–49	32 (15.8%)	19 (20.0%)
	50–59	21 (10.4%)	27 (28.4%)
	>60	16 (7.9%)	14 (14.7%)
Education**	None	(0.0%)	3 (3.2%)
	Primary	7 (3.5%)	41 (43.2%)
	Secondary	94 (46.5%)	41 (43.2%)
	Technical	55 (27.2%)	(0.0%)
	University	46 (22.8%)	10 (10.5%)
Occupation**	Housewife	82 (40.6%)	16 (16.8%)
	Student	22 (10.9%)	9 (9.5%)
	Sales	20 (9.9%)	18 (19.0%)
	Health/Education	21 (10.4%)	2 (2.1%)
	Skilled Work	7 (3.5%)	10 (10.5%)
	Police/Military	1 (0.5%)	3 (3.2%)
	Unqualified	14 (6.9%)	8 (8.4%)
	Driver	1 (0.5%)	4 (4.2%)
	Office work	7 (3.4%)	12 (12.6%)
	Other	27 (13.4%)	13 (13.7%)
RDT Tested on	Self-Administered	115 (56.9%)	50 (52.6%)
	Adult Family Member	50 (24.8%)	33 (34.7%)
	Child Family Member	29 (14.4%)	1 (1.1%)
	Friend/Other	8 (4.0%)	11 (11.6%)

<sup>\*</sup>p-value = 0.0349

https://doi.org/10.1371/journal.pntd.0009307.t001

<sup>\*\*</sup>p-value<0.0001; for age the 18–29 y/o category was compared to  $\geq$ 30 years of age; for education categories were collapsed to none/primary, secondary, and university/technical; for occupation categories were housewives versus other professions.

Table 2. Observations of community members' competency applying two rapid diagnostic tests (two-line, five-line) in Iquitos, Peru and Phnom Penh, Cambodia. The number of participants scored in each competency category, high = 3, medium (med) = 2, and poor = 1, are shown for each testing step.

Testing step (Two-line test)	Level of competence observed when applying RDT								
	Peru (n = 101)			Cambodia (n = 42)					
	Poor	Med	High	%High	Poor	Med	High	% High	
Recognizes materials	0	20	81	80.2	4	27	11	26.2	
Locates puncture site	0	16	85	84.2	5	7	30	71.4	
Cleans site	0	6	95	94.1	5	8	29	69.1	
Lancet use	1	30	70	69.3	11	4	17	40.5	
Capillary tube use	0	18	83	82.2	11	12	19	45.2	
Blood collection	1	15	85	84.2	9	11	22	52.4	
Locates well	0	8	93	92.1	4	8	30	71.4	
Blood transfer	0	9	92	91.1	5	5	32	76.2	
Adds buffer	0	5	96	95.0	7	13	22	52.4	
Waits for result	2	4	95	94.1	4	10	28	66.7	
Reads result	0	9	92	91.1	8	7	27	64.3	
Marks result on paper	1	6	94	93.1	8	9	25	59.5	
Testing step (Five-line test)	Peru (n = 101)			Cambodia (n = 53)					
	Poor	Med	High	% High	Poor	Med	High	% High	
Recognizes materials	0	37	64	63.3	5	36	12	22.6	
Add reagent to tube	0	28	73	72.3	-	-	_	_	
Locates puncture site	1	28	72	71.3	6	17	30	56.6	
Cleans site	1	18	82	81.2	2	17	34	64.2	
Lancet use	0	36	65	64.4	14	20	19	35.9	
Capillary tube use	1	42	58	57.4	20	21	12	22.6	
Blood collection	5	47	49	48.5	8	26	19	35.8	
Blood transfer to tube	1	21	79	78.2	_	-	_		
Mix blood in tube	1	31	69	68.3	-	-	_	_	
Locates well	0	20	81	80.2	11	14	28	52.8	
Blood transfer	0	19	82	81.2	8	22	23	43.4	
Adds buffer	0	12	89	88.1	9	16	28	52.8	
Waits for result	0	10	91	90.1	5	10	38	71.7	
Reads result	0	12	89	88.1	7	9	37	69.8	
Marks result on paper	2	14	85	84.2	10	14	29	54.7	

https://doi.org/10.1371/journal.pntd.0009307.t002

Education levels, however, were higher overall in Peru than Cambodia ( $\chi 2 = 96.2$ , df = 2; p<0.0001). Forty-one percent of the participants were housewives in Peru compared to 17% in Cambodia ( $\chi 2 = 16.5$ , df = 1; p<0.0001), where most of the participants were low-skilled laborers. The majority of RDTs were self-administered in both countries, but when administered on others, they were done on family members and friends, including young children (as low as 5 years of age).

In Iquitos, >80% (range 80–95%) of residents testing the two-line test were able to perform 11 out of 12 necessary steps competently without requesting help (scored a '3' [high in Table 2] by a trained observer). The most challenging step was using the retractable lancet (69% used the lancet correctly without help) (see Table 2). The steps associated with starting the procedure (recognizing test materials) and blood collection (locate puncture site, capillary tube use, blood collection) required more help than the remaining steps. The five-line test, with additional steps, was more difficult for residents than the two-line test. High competency was observed for >80% (range 80–91%) for only 7 of 15 steps evaluated.

In Phnom Penh, users universally struggled to perform the required tasks (every category, both devices). The percentage of participants demonstrating high competency on their ability to execute the steps appropriately without asking for help, ranged from 26–76% and 23–72%, for the two-line and five-line tests, respectively. As in Iquitos, Phnom Penh participants had more problems with the initial steps of the procedures, specifically, recognition of the test kit materials and blood collection steps (Table 2). Although CHWs tended to do slightly better than community members, the overall difference was not statistically significant, so the aggregated results are presented in Table 2.

To assess factors contributing to correct RDT use, we developed an overall "competency" score for each participant's use of one of the two RDTs. For each of the steps observed, the scores were summed and divided by the number of steps assessed, providing a final competency score between one and three. The competency scores were higher overall in Peru, ranging from 2.00 to 3.00 (IQR 2.75–3.00), than Cambodia where scores ranged from 1.08 to 3.00 (IQR 1.92–2.75). To examine factors associated with competency, we created a dichotomous variable for competency: high (competency score >2.9) and low to medium competency score  $\le$  2.9).

In, Iquitos household members performed the two-line test significantly better than the five-line test; the odds of a high competency score were 4.00 (95% CI 2.13–7.54) for the two-line test compared to the five-line test. Younger household residents (18–34 years of age) were 2.28 (95% CI 1.21–4.28) times more likely to perform the test at a high level than older household residents (34–78 years of age). Education level was also an important factor; people reporting any post-secondary education (university or technical school) were 2.69 times (95% CI 1.45–5.00) more likely to perform the test at a higher competency level than participants reporting primary or secondary education only.

In Cambodia, no significant difference in ability to perform the test was observed between the two devices, between CHWs and householders, or by age. Only education level had a statistically significant effect: participants with post-secondary education were 12.49 times (95% CI 1.17–133) more likely to perform the test at a high competency level compared to participants with less education (none, primary, or secondary).

To assess how many repetitions of the test would be necessary to perform the test competently, eight CHWs (different from those who participated in the earlier study) carried out the two-line (n = 4) and five-line (n = 4) tests up to six times on different individuals. The only feedback provided was during the testing process, when the observer intervened if requested or a significant mistake was imminent. As expected, the learning curve was fast; by the sixth trial, all four CHWs using the two-line RDT had perfect scores in all steps, without needing any help. The steps that required the most practice were waiting for results (one took five attempts before doing this correctly), reading results, capillary tube use, and blood collection (three CHWs took four attempts before doing these steps correctly). Scores also dramatically improved for the five-line device, although one CHW still struggled with reading and interpreting the results even at the sixth attempt. Additional steps that required various attempts prior to achieving perfect scores were blood collection, blood transfer, adding buffer, and waiting for results.

# Use of commercially available RDTs by community members with febrile illness (Peru only)

In Iquitos, we provided FDA approved and commercially available SD Bioline Dengue Duo test kits (NS1 test), including all the items needed for their use and written and video instructions, to 74 households participating in 56 cluster investigations [14]. Seventy-four (72%) of

the 103 houses consented to participate. Of these 74 households, 17 of the clusters were initiated by a dengue case, and 39 clusters were initiated by a febrile case that was negative for dengue but with a presumptive or laboratory confirmed Zika case. When Zika virus was circulating (June 2016 to April 2017), we explained to the participants that there was still a possibility that any febrile illness could be dengue, but that the test might also be positive for Zika because of cross reactivity [24]. Within participating households, 293 of 445 (66%) residents provided consent. Out of these 74 households and 293 individuals consented, there were 14 reports of individuals who developed dengue-like symptoms, from 10 households, that could have used the NS1 test. Of these, seven of the NS1 tests were applied to six individuals: one on an adult, five on children between the ages of three and 13, and in one participant, a second test was repeated by the family member. Eight individuals who could have used the available test in their home decided not to use it.

The primary reasons for not using the NS1 test were fear of performing the test incorrectly or "wasting it", or they reported a lack of fever with their other symptoms so they assumed it was Zika and should not use the test. They preferred to wait for the phlebotomist who was part of our research team to test them rather than self-administer the test. One individual forgot that they had the test altogether.

#### **Discussion**

Our study provides evidence that home-based RDT use (with RDTs of limited complexity) may be feasible in certain communities, but less so in others, depending on the demographic and educational characteristics of the populations that would use them. Additional research on the necessity of training and locally adapted RDT implementation strategies are needed before their incorporation into public health programs. For example, in our Cambodian site, neither community members or CHWs were able to apply the RDT tests with sufficient competency to justify their deployment in this site without training, but repeated use among CHWs improved performance suggesting that with training, use by CHWs (and possibly community members) could be feasible.

In locations where CHW programs are in place, the advantages of providing formal training to volunteers who are already trusted by their community and have established connections with the health care system could outweigh any of the perceived benefits from home based RDT use–and this is what is done successfully for malaria RDT programs, for example [25]. CHWs could easily receive dedicated RDT training with refreshers, carry and apply RDTs regularly, interpret tests correctly, and ensure that a patient receives proper follow-up depending on the test results. Overall, there is considerable evidence that CHWs improve health outcomes and reduce health disparities, and when appropriate, should be promoted in communities around the world, particularly in areas where infectious diseases are endemic [26].

CHW programs, however, are not universally available, as in Iquitos, and there are circumstances where home-based testing may be more appropriate or justified. For example, for diseases with a stigma (i.e., HIV and other sexually transmitted diseases) or associated with high mortality or are highly contagious (i.e., COVID-19 and Ebola), home-based RDT strategies without training may be necessary and the most appropriate option to address privacy and/or safety concerns. Our study directly tested the feasibility of home-based testing of two prototype devices without the benefit of training.

Competency was higher for the less complex two-line test compared to the five-line test in both countries, but this difference was only statistically significant in the Peru site, where performance of community members was sufficiently high to consider future deployments.

Overall, "lancet use," "capillary use," and "blood collection" were the categories in which participants had the most difficulties. For example, for the two-line test, only 69% and 43% of the participants were able to use a retractable lancet correctly in Iquitos and Phnom Penh, respectively. The challenge of extracting blood by lancet has previously been reported for in-home HIV RDTs [27]; considerations about types of lancets and clear instructions are necessary to make it more user-friendly. The five-line test, performance was worse because of the additional steps of collecting the blood from the finger, mixing it with a buffer, and then using a different pipette to transfer this blood and buffer to the RDT. Home-based RDTs should be designed to have as few simple steps as possible, and specific devices must be tested before wider deployment. Interest in multiplex tests is attractive but needs to be balanced with simplicity for deployment to remote sites.

This study found that the enthusiasm expressed by study participants from Iquitos for using "dengue tests" was consistent with that observed in prior focus group discussions carried out in both sites [9]. During the current study, some of our Peruvian participants expressed a sense of empowerment using the RDTs. Despite their initial fears, they were pleasantly surprised and proud of their performance. Our study nurse noted that small children were willingly let their mother stick their finger even though they would cry during the procedure at hospital or health facility. Also, in Iquitos, competency did not appear to be influenced by having prior exposure to participating in a research project, as we recruited study participants evenly from neighborhoods with and without ongoing research studies.

Instructional and training strategies are key issues to examine in the context of future implementation. The first step would be to conduct a rigorous site-specific validation of instructional materials. Circumstances in each study site resulted in different instructional materials for RDT users—including the availability of a video in Iquitos. The role of the video instructions in the observed differences in competency between Iquitos and Phnom Penh warrant further investigation: did the video alone make a big difference between sites? With the explosion of "YouTube" instructional videos, and in the context of possible infectious disease for which isolation is important, the value of providing this type of medium for instructions is possible. Future studies should include an evaluation of different types and levels of instructional materials with and without the inclusion of formal training.

Also critical is the identification of what would motivate home or community based RDT use in diverse settings. In Peru, it was the possibility of streamlining services once the patient reached health facilities [9]. An RDT dengue result would be useful to a physician but clinical follow up should be sought independent of the test result. Our participants were hopeful that a preliminary test would reduce the wait time and number of steps once seen at a health facility [9]. There are many potential programmatic benefits of home-based RDT testing for dengue virus infection if this testing could be incorporated into government surveillance or outbreak response. One relevant application could be for locations where COVID-19 and dengue are co-circulating and cause a case management conundrum. For example, in Iquitos, prior to the COVID-19 pandemic, suspected dengue cases were encouraged to present to health care facilities for clinical monitoring, a practice that should now be discouraged unless patients have dengue warning signs [28]. In these current circumstances, a resident with a dengue positive test result could potentially be directed to a dengue specific facility. Coordinating telemedicine strategies with home RDTs could potentially be used for triage of dengue and COVID-19 patients, reducing contact at health facilities. Policies surrounding home RDTs are yet to be discussed on the large-scale but could provide innovative strategies for navigating the COVID-19 emergency, or others in the future.

A principal limitation of this study was the ability compare RDT competency between Peru and Cambodia, because the two sites had different instructional methods (written/graphical

information versus video), implementation strategies (community health workers versus community members), and contexts (culturally, geographically, and even types of diseases and health systems). Training of the personnel carrying out the observations was done by country specific teams with the same reporting format, but we recognize that some scoring differences were possible. For this reason, we have avoided direct comparisons between the two countries when possible since this was not the focus of the study.

Another important study limitation was determining if Iquitos residents would use an RDT if they were provided at no cost. Dengue transmission rates were low during the study period, resulting in an inadequate sample size to draw meaningful conclusions; there were only 14 febrile participants with access to home RDT testing, of which six used the RDT. Our contact cluster study design maximized the possibility of finding febrile contacts, but these households had constant contact with study personnel which may have biased residents' decision to use them. Regular visits from our staff reminded people to use the tests, but was also one of the most frequently cited reasons not to use the tests (why use the test when someone trained would do it the next day?) Also, complicating this study was the requirement to have informed consent ahead of time. All these factors made it difficult to test home-based RDT use under real-life circumstances.

In conclusion, we found that in Peru, using a simple device, a majority of community members using instructions, but no training, were able to competently use an RDT on themselves or someone else. RDT competency, however, was not achieved for the device with more complex steps or for either device in Cambodia. Further research is needed to determine the level and type of instructions needed that is locally appropriate contrasted with training of key community members (volunteers, pharmacy workers, school nurses, or health promoters), and how to best integrate RDTs into different health systems before use or scale up. The potential applications of home-based RDT use are numerous, in particular, assisting in managing outbreaks, especially when there is a need to reduce contact with health facilities to a minimum as observed during the COVID-19 pandemic.

#### **Supporting information**

S1 Text. Written Instructions provided to community or health workers (Cambodia only) to perform self-testing.
(PDF)

S2 Text. Video instructions provided to community members on DVD to perform self-testing for two-line (https://youtu.be/5RQtc9KNA5A), five-line (https://youtu.be/adv-yQaI\_iA) used for observational competency studies and the SDBioline Dengue Duo https://youtu.be/arOSOaeulIw.

(DOCX)

S3 Text. Observational data collection sheet used by study observers. (DOCX)

S1 Raw data. Raw data set (Raw\_data.xlsx) used for Tables 1 and 2 and competency score analyses.

(XLSX)

#### **Acknowledgments**

We would like to thank the residents of Iquitos for their support and participation in this study. We greatly appreciate support of the Loreto Regional Health Department including

Drs. Hugo Rodriguez-Ferruci, Christian Carey, Carlos Alvarez, Hernan Silva and the Lic. Wilma Casanova Rojas who all facilitated our work in Iquitos. Drs. Eric Halsey and Christopher Mores provided institutional support from the U.S. Naval Medical Research Unit No. 6. A special thanks to Gloria Talledo for her ongoing support with the preparation of IRB protocols and reports for this project. We also appreciate the careful commentary and advice provided by the NAMRU-6 Institutional Review Board and Research Administration Program for the duration of this study.

Copyright statement: Author Robert D. Hontz is a U.S. military service member. This work was prepared as part of his official duties. Title 17 U.S.C. § 105 provides that 'Copyright protection under this title is not available for any work of the United States Government'. Title 17 U. S.C. § 101 defines a U.S. Government work as a work prepared by military service members or employees of the U.S. Government as part of those persons' official duties.

The views expressed in this article are those of the authors and do not necessarily reflect the official policy or position of the Department of the Navy, Department of Defense, nor the U.S. Government.

#### **Author Contributions**

**Conceptualization:** Amy C. Morrison, Chhorvann Chhea, Vonthanak Saphonn, Robert D. Hontz, Pamina M. Gorbach, Valerie A. Paz-Soldan.

Data curation: Amy C. Morrison, Jhonny Cordova, Heng Sopheab.

**Formal analysis:** Amy C. Morrison, Jennie L. Mckenney, Vonthanak Saphonn, Pamina M. Gorbach, Valerie A. Paz-Soldan.

Funding acquisition: Amy C. Morrison, Robert D. Hontz, Valerie A. Paz-Soldan.

Investigation: Amy C. Morrison, Jennie L. Mckenney, Jhonny Cordova, Jennifer E. Rios, W. Lorena Quiroz, S. Alfonso Vizcarra, Heng Sopheab, Chhorvann Chhea, Vonthanak Saphonn, Pamina M. Gorbach, Valerie A. Paz-Soldan.

**Methodology:** Amy C. Morrison, Heng Sopheab, Chhorvann Chhea, Vonthanak Saphonn, Robert D. Hontz, Pamina M. Gorbach, Valerie A. Paz-Soldan.

**Project administration:** Amy C. Morrison, Robert D. Hontz, Pamina M. Gorbach, Valerie A. Paz-Soldan.

**Supervision:** Amy C. Morrison, Robert D. Hontz, Pamina M. Gorbach, Valerie A. Paz-Soldan. **Validation:** Heng Sopheab.

**Writing – original draft:** Amy C. Morrison, Julia Schwarz, Jennie L. Mckenney, Karin M. Bauer

Writing – review & editing: Amy C. Morrison, Chhorvann Chhea, Robert D. Hontz, Pamina M. Gorbach, Valerie A. Paz-Soldan.

#### References

- Pfützner A. Advances in patient self-monitoring of blood glucose. J Diabetes Sci Technol. 2016. Available: https://journals.sagepub.com/doi/abs/10.1177/1932296815619183.
- Doshi ML. Accuracy of consumer performed in-home tests for early pregnancy detection. Am J Public Health. 1986; 76: 512–514. https://doi.org/10.2105/ajph.76.5.512 PMID: 3963279
- 3. Mabey DC, Sollis KA, Kelly HA, Benzaken AS, Bitarakwate E, Changalucha J, et al. Point-of-care tests to strengthen health systems and save newborn lives: the case of syphilis. PLoS Med. 2012; 9: e1001233. https://doi.org/10.1371/journal.pmed.1001233 PMID: 22719229

- Visser T, Daily J, Hotte N, Dolkart C, Cunningham J, Yadav P. Rapid diagnostic tests for malaria. Bull World Health Organ. 2015; 93: 862–866. https://doi.org/10.2471/BLT.14.151167 PMID: 26668438
- Kristensen GBB, Nerhus K, Thue G. Standardized evaluation of instruments for self-monitoring of blood glucose by patients and a technologist. Clinical. 2004. Available: http://clinchem.aaccjnls.org/content/ 50/6/1068.short. https://doi.org/10.1373/clinchem.2004.031575 PMID: 15161725
- Jelinek T, Amsler L, Grobusch MP, Nothdurft HD. Self-use of rapid tests for malaria diagnosis by tourists. The Lancet. 1999. p. 1609. https://doi.org/10.1016/s0140-6736(99)01969-8 PMID: 10560678
- Trachsler M, Schlagenhauf P, Steffen R. Feasibility of a rapid dipstick antigen-capture assay for self-testing of travellers' malaria. Trop Med Int Health. 1999; 4: 442–447. <a href="https://doi.org/10.1046/j.1365-3156.1999.00419.x">https://doi.org/10.1046/j.1365-3156.1999.00419.x</a> PMID: 10444320
- Lee VJ, Tan SC, Earnest A, Seong PS, Tan HH, Leo YS. User acceptability and feasibility of self-testing with HIV rapid tests. J Acquir Immune Defic Syndr. 2007; 45: 449–453. <a href="https://doi.org/10.1097/QAI.0b013e318095a3f3">https://doi.org/10.1097/QAI.0b013e318095a3f3</a> PMID: 17554213
- Paz-Soldan VA, Morrison AC, Sopheab H, Schwarz J, Bauer KM, Mckenney JL, et al. Potential Use of Community-Based Rapid Diagnostic Tests for Febrile Illnesses: Formative Research in Peru and Cambodia. PLoS Negl Trop Dis. 2019; 13: e0007773. https://doi.org/10.1371/journal.pntd.0007773 PMID: 31658252
- Morrison AC, Schwarz J, Long KC, Cordova J, Rios JE, Quiroz WL, et al. Acceptability of Aedes aegypti blood feeding on dengue virus-infected human volunteers for vector competence studies in Iquitos, Peru. PLoS Negl Trop Dis. 2019; 13: e0007090. <a href="https://doi.org/10.1371/journal.pntd.0007090">https://doi.org/10.1371/journal.pntd.0007090</a> PMID: 30742621
- Aramburú Guarda J, Ramal Asayag C, Witzig R. Malaria reemergence in the Peruvian Amazon region. Emerg Infect Dis. 1999; 5: 209–215. https://doi.org/10.3201/eid0502.990204 PMID: 10221872
- 12. Durand Velazco S, Fiestas Solórzano V, Sihuincha Maldonado M, Chávez Lencinas C, Vásquez Vela V, Torrejón Flores C, et al. [Impact of the dengue epidemic due to a new lineage of DENV-2 American/ Asian genotype in the health services demand in hospital "Cesar Garayar Garcia", Iquitos]. Rev Peru Med Exp Salud Publica. 2011; 28: 157–159. https://doi.org/10.1590/s1726-46342011000100027 PMID: 21537788
- Paz-Soldan VA, Plasai V, Morrison AC, Rios-Lopez EJ, Guedez-Gonzales S, Grieco JP, et al. Initial
  assessment of the acceptability of a Push-Pull Aedes aegypti control strategy in Iquitos, Peru and Kanchanaburi, Thailand. Am J Trop Med Hyg. 2011; 84: 208–217. <a href="https://doi.org/10.4269/ajtmh.2011.09-0615">https://doi.org/10.4269/ajtmh.2011.09-0615</a> PMID: 21292886
- Stoddard ST, Forshey BM, Morrison AC, Paz-Soldan VA, Vazquez-Prokopec GM, Astete H, et al. House-to-house human movement drives dengue virus transmission. Proc Natl Acad Sci U S A. 2013; 110: 994–999. https://doi.org/10.1073/pnas.1213349110 PMID: 23277539
- Getis A, Morrison AC, Gray K, Scott TW. Characteristics of the spatial pattern of the dengue vector, Aedes aegypti, in Iquitos, Peru. Am J Trop Med Hyg. 2003; 69: 494–505. PMID: 14695086
- Morrison AC, Minnick SL, Rocha C, Forshey BM, Stoddard ST, Getis A, et al. Epidemiology of dengue virus in Iquitos, Peru 1999 to 2005: interepidemic and epidemic patterns of transmission. PLoS Negl Trop Dis. 2010; 4: e670. https://doi.org/10.1371/journal.pntd.0000670 PMID: 20454609
- Rocha C, Morrison AC, Forshey BM, Blair PJ, Olson JG, Stancil JD, et al. Comparison of two active surveillance programs for the detection of clinical dengue cases in Iquitos, Peru. Am J Trop Med Hyg. 2009; 80: 656–660. PMID: 19346395
- Schaber KL, Paz-Soldan VA, Morrison AC, Elson WHD, Rothman AL, Mores CN, et al. Dengue illness impacts daily human mobility patterns in Iquitos, Peru. PLoS Negl Trop Dis. 2019; 13: e0007756. https://doi.org/10.1371/journal.pntd.0007756 PMID: 31545804
- Elson WH, Reiner RC, Siles C, Bazan I, Vilcarromero S, Riley-Powell AR, et al. Heterogeneity of dengue illness in community-based prospective study, Iquitos, Peru. Emerg Infect Dis. 2020; 26: 2077–2086. https://doi.org/10.3201/eid2609.191472 PMID: 32818402
- Mulvaney SP, Fitzgerald LA, Hamdan LJ, Ringeisen BR, Petersen ER, Compton JR, et al. Rapid design
  and fielding of four diagnostic technologies in Sierra Leone, Thailand, Peru, and Australia: Successes
  and challenges faced introducing these biosensors. Sensing and Bio-Sensing Research. 2018; 20: 22
  33.
- Hunsperger EA, Yoksan S, Buchy P, Nguyen VC, Sekaran SD, Enria DA, et al. Evaluation of commercially available diagnostic tests for the detection of dengue virus NS1 antigen and anti-dengue virus IgM antibody. PLoS Negl Trop Dis. 2014; 8: e3171. <a href="https://doi.org/10.1371/journal.pntd.0003171">https://doi.org/10.1371/journal.pntd.0003171</a> PMID: 25330157
- Long KC, Sulca J, Bazan I, Astete H, Jaba HL, Siles C, et al. Feasibility of feeding Aedes aegypti mosquitoes on dengue virus-infected human volunteers for vector competence studies in Iquitos, Peru.

- PLoS Negl Trop Dis. 2019; 13: e0007116. https://doi.org/10.1371/journal.pntd.0007116 PMID: 30753180
- Paz-Soldan VA, Reiner RC Jr, Morrison AC, Stoddard ST, Kitron U, Scott TW, et al. Strengths and weaknesses of Global Positioning System (GPS) data-loggers and semi-structured interviews for capturing fine-scale human mobility: findings from Iquitos, Peru. PLoS Negl Trop Dis. 2014; 8: e2888. https://doi.org/10.1371/journal.pntd.0002888 PMID: 24922530
- 24. Wen J, Shresta S. Antigenic cross-reactivity between Zika and dengue viruses: is it time to develop a universal vaccine? Curr Opin Immunol. 2019; 59: 1–8. <a href="https://doi.org/10.1016/j.coi.2019.02.001">https://doi.org/10.1016/j.coi.2019.02.001</a> PMID: 30884384
- Counihan H, Harvey SA, Sekeseke-Chinyama M, Hamainza B, Banda R, Malambo T, et al. Community health workers use malaria rapid diagnostic tests (RDTs) safely and accurately: results of a longitudinal study in Zambia. Am J Trop Med Hyg. 2012; 87: 57–63. <a href="https://doi.org/10.4269/ajtmh.2012.11-0800">https://doi.org/10.4269/ajtmh.2012.11-0800</a>
   PMID: 22764292
- Cosgrove S, Moore-Monroy M, Jenkins C, Castillo SR, Williams C, Parris E, et al. Community health workers as an integral strategy in the REACH U.S. program to eliminate health inequities. Health Promot Pract. 2014; 15: 795–802. https://doi.org/10.1177/1524839914541442 PMID: 25063590
- Ibitoye M, Frasca T, Giguere R, Carballo-Diéguez A. Home testing past, present and future: lessons learned and implications for HIV home tests. AIDS Behav. 2014; 18: 933–949. <a href="https://doi.org/10.1007/s10461-013-0668-9">https://doi.org/10.1007/s10461-013-0668-9</a> PMID: 24281697
- WHO/TDR. Dengue: guidelines for diagnosis, treatment, prevention and control—New edition. 2009. p. 147.